Author's response to reviews

Title: The contribution of psychological distress to socio-economic differences in cause-specific mortality: a population-based follow-up of 28 years.

Authors:

Kirsi M Talala (kirsi.talala@thl.fi)
Taina M Huurre (taina.huurre@thl.fi)
Tiina KM Laatikainen (tiina.laatikainen@thl.fi)
Tuija P Martelin (tuija.martelin@thl.fi)
Aini I Ostamo (aini.ostamo@thl.fi)
Ritva S Prättälä (ritva.prattala@thl.fi)

Version: 4 Date: 24 November 2010

Author's response to reviews: see over
Helsinki, 24th November 2010

The BMC Series Editorial Production Team

Dear Editorial Production Team,

We would like to submit our revised manuscript (MS:3759071194339361) “The contribution of psychological distress to socio-economic differences in cause-specific mortality: a population-based follow-up of 28 years”. The manuscript has been corrected according to the changes suggested by the editorial production team. The details of revisions are listed in this letter further below. We hope you will find these revisions satisfactory.

Sincerely yours,
Kirsi Talala

Kirsi Talala, Researcher, M.Soc.Sc,
National Institute for Health and Welfare (THL),
P.O. Box 30, FI-00271 Helsinki, Finland
mobile +358-40-7654085
e-mail: kirsi.talala@thl.fi
Author list: Kirsi M Talala, Taina M Huurre, Tiina KM Laatikainen, Tuija P Martelin, Aini I Ostamo and Ritva S Prättälä

Title: The contribution of psychological distress to socio-economic differences in cause-specific mortality: a population-based follow-up of 28 years.

Journal: BMC Public Health

MS ID: 3759071194339361

Point-by-point description of the changes made

Tables: The first tables referenced in the manuscript text are labelled Tables 1-3, instead of 2-4 (page 9). The numbering of the tables is revised throughout the manuscript (pages 10-12). Tables are provided in the new order at the end of the manuscript. A title for Table 4 was moved to the same page with the table as it was on a previous page. Additionally table placement texts were removed, on page 11 and 12.

Typography: Following corrections were made in the reference list, on page 17:
- reference 1, 2, 5, all the titles were revised having the similar form of ‘EU Working Group on Socioeconomic Inequalities in Health’
- reference 14; the name of the journal was abbreviated as ‘Eur J Population’
- reference 30; the name of the journal was abbreviated as ‘Soc Indic Res’